Author's response to reviews

Title: Distraction osteogenesis in a severe mandibular deficiency

Authors:

Kerim Ortakoglu (kortakoglu@yahoo.com)
Seniz Karacay (senkaracay@yahoo.com)
Metin Sencimen (metinsencimen@yahoo.com)
Erol Akin (erol1akin@hotmail.com)
Aykut H Ozyigit (aykutozyigit@yahoo.com)
Osman Bengi (osmanbengi@hotmail.com)

Version: 4 Date: 5 December 2006

Author's response to reviews:

Figures 1,5,6,14,18,19 are deleted

Figures 2,3 are cropped

Figure 13 is replaced with a clear focused one

Figure legends are rearranged